Author's response to reviews

Title: Measuring the tail of the dog that doesn't bark in the night: the case of the national evaluation of Choose Life (The National Strategy and Action Plan to Prevent Suicide in Scotland)

Authors:

Mhairi Mackenzie (m.mackenzie@clinmed.gla.ac.uk)
Avril Blamey (avril.blamey@health.scot.nhs.uk)
Emma Halliday (emma.halliday@health.scot.nhs.uk)
Margaret Maxwell (margaret.maxwell@ed.ac.uk)
Allyson McCollam (allyson@sdcmh.org.uk)
David McDaid (d.mcdaid@lse.ac.uk)
Joanne McLean (joanne@sdcmh.org.uk)
Amy Woodhouse (amy@sdcmh.org.uk)
Steve Platt (steve.platt@ed.ac.uk)

Version: 3 Date: 21 June 2007

Author's response to reviews:

Dear Editorial team

I have made most of the changes as requested to the above manuscript however there are two changes that I have been unable to make - these are to the abstract and manuscript structure. The paper can't be structured as recommended because it does not report on findings - it is a reflective paper on the challenges of evaluating rather than the results of an evaluation. (For this reason it has been designated as a correspondence article). I attach the manuscript and will now submit formally online.

The changes made are: correction of authors list, removal of qualifications, addition of countries in affiliations, underlinings removed, references in normal size font, removal of figure and table placement text, chang of box 1 to table 1 and reordering of later sections of manuscript.

I hope that this is satisfactory

Best wishes

Mhairi Mackenzie